

Author's response to reviews

Title: Effects Of Testicular Microlithiasis On Doppler Parameters: Report Of Three Cases

Authors:

Ramazan Kutlu(rkutlu@inonu.edu.tr)
Ahmet Sigirci(asigirci@inonu.edu.tr)
Tamer Baysal(tbaysal@inonu.edu.tr)
Alpay Alkan(aalkan@inonu.edu.tr)
Kaya Sarac(ksarac@inonu.edu.tr)

Version:6 **Date:**6Feb2002

Dear Editor,

I would like to thank you for interest in our manuscript.

We have made the following changes:

1. We accept the reviewer's edited version of our manuscript and resubmitted the final version.
2. References were changed into the BioMed style.
3. We have no competing interests and a statement was added to the title page.
4. All authors linked to their corresponding affiliation using superscript numerals.
5. Although in checklist number 13 it is stated that "Abstracts must be structured into Background, Methods, Results, Conclusions", we think this is not applicable to our manuscript (there was no suggestion from reviewer for this issue). If it is needed, could you please inform us?
6. The identities of the patient were not revealed both in the manuscript and figures. If it is necessary we can provide consent of them.

Best regards,

Ramazan Kutlu, MD
rkutlu@inonu.edu.tr